Author’s response to reviews

Title: Measuring the effectiveness of an intensive IPV training program offered to Greek general practitioners and residents of general practice.

Authors:

 Maria Papadakaki (mpapadakaki@yahoo.gr)
 Eleni Petridou (epetrid@med.uoa.gr)
 Manolis Kogevas (kogevas@creal.cat)
 Christos Lionis (lionis@galinos.med.uog.gr)

Version: 3 Date: 6 March 2013

Author's response to reviews:

Title: Measuring the effectiveness of an intensive IPV training program offered to Greek general practitioners and residents of general practice.

We would like to thank both the reviewers for their second review. All the changes introduced in the manuscript are shown in red colour and a point-by-point response is provided below.

Referee 2

Comment 1
Contents of training program, number 2, should read 'Dynamics of IPV'

Response 2
Thanks for indicating this mistyping. In Appendix I, the sentence now reads “Dynamics of IPV” instead of “Dynamics of PV”.

Comment 2
There is some problem with the presentation of PREMIS section C, in terms of the instruction in parentheses for the first item on strongest single risk factor. The content here needs to appear as an instruction for the whole of this section surely, rather than related to the first item.

Response 2
Thanks for noticing this. In Appendix II and particularly in Section C, instruction in parentheses was separated for the first item.

Referee 1

Comment 1
Please rewrite the discussion to more accurately reflect the limitations of your findings and design. Your opening paragraph does not match these, but your later one do.
Response 1:
Many thanks for this important comment. Certain sentences of the first paragraph of the discussion have been rephrased to avoid generalization of findings and better reflect the limitations of the study.

General comment of both reviewers:
Needs some language corrections before being published

Response:
The manuscript has been thoroughly checked for language mistakes. Language corrections are shown in red colour.